Author's response to reviews

Title: A Randomized Controlled Trial to Prevent Glycemic Relapse in Longitudinal Diabetes Care: Study Protocol (NCT00362193)

Authors:

Mary Margaret Huizinga (mary.margaret.huizinga@vanderbilt.edu)
Ayumi Shintani (ayumi.shintani@vanderbilt.edu)
Stephanie Michon (stephanie.michon@vanderbilt.edu)
Anne Brown (anne.brown@vanderbilt.edu)
Kathleen Wolff (kathleen.wolff@vanderbilt.edu)
Laurie Shackleford (laura.shackleford@vanderbilt.edu)
Elaine Boswell-King (elaine.boswell.king@vanderbilt.edu)
Rebecca PRATT Gregory (becky.gregory@vanderbilt.edu)
Dianne Davis (dianne.davis@vanderbilt.edu)
Renee A Stiles (renee.stiles@vanderbilt.edu)
Tebeb Gebretsadik (tebeb.gebretsadik@vanderbilt.edu)
Kong Chen (kong.chen@vanderbilt.edu)
Russell L Rothman (russell.rothman@vanderbilt.edu)
James W Pichert (jim.pichert@vanderbilt.edu)
David Schlundt (david.schlundt@vanderbilt.edu)
Tom A Elasy (tom.elasy@vanderbilt.edu)

Version: 2 Date: 17 October 2006

Author's response to reviews: see over
October 17, 2006

Editor  
*Implementation Science*

To the Editor:

Please consider the revised manuscript, “Longitudinal Diabetes Care: A Randomized Controlled Trial to Prevent Glycemic Relapse”, for publication as an original article in *Implementation Science*.

We have revised this manuscript to reflect the *Implementation Science* draft study protocol guidelines, the CONSORT guidelines and the Ottawa statement. The major changes are reflected in an updated title and additions to the abstract and the study methods. The title of the manuscript now includes the trial registry identification number and the words “study protocol.” The abstract and the methods have been updated to include the study setting, more information about the randomization and blinding of the participants and study coordinators.

Thank you for the opportunity to revise this manuscript. We have attempted to be responsive to the major concerns outlined. We hope that you will find this revision acceptable but stand ready to further modify should you feel that is warranted. Again, thank you.

Sincerely,

Mary Margaret Huizinga, MD  
VA Tennessee Valley Healthcare System  
GRECC 1310 24th Avenue South  
Nashville, TN 37210  
Phone: 615-327-4751, ext 6817; Fax: 615-327-5381  
Email: mary.margaret.huizinga@vanderbilt.edu

Corresponding author:  
Tom A Elasy, MD, MPH  
Vanderbilt University Medical Center  
8210 MCE, South Tower  
Nashville, TN 37232-8148  
Phone:615-936-7772; Fax: 615-936-8077  
Email: tom.elasy@vanderbilt.edu